Updated sensor status

Here is a revised table, now that some things have been fixed (and others have died):

Site

2 [network died
last night]

3

4 ?[network
resetting]

5

Battery
&Solar

ok

ok

still lowest

ok

ok

ok
ok

Sonic

head off

ok

ok

ok

ok

ok
ok

Krypton

not installed

not installed

not installed

not installed

not installed

not installed

not installed

ok

ok

ok

ok

ok

ok
ok

Baro

TRH

not installed

not installed

not installed

not installed

ok (reprogrammed
sensor installed)

ok

not installed

Rad

Rsw.in/Rsw.out need
new coefficients
Riw ok

Rsw.out needs new coef
Rsw.in/Rlw ok

ok

ok

4-comp flipped
yesterday

Rsw.in needs new coef
Rsw.out/Rlw ok

ok
ok

Soil

ok

ok

turned off?

Tsoil 0.6/3.1/4.

4 all bad
ok

Tsoil dead
ok

Soil.
aux

mote died
last night

not planned

not planned

not planned

Tsoil.3.1/4.
4cm bad
Qsoil erratic
Gsoil ok
TPO1 not
reporting

ok

not planned

Precip

not
planned

not
planned

no fluid

ok?

not
planned

no fluid

not
planned

Shadowband

SPN1 reported ~1.0/0.3
for a day, then died
Licor not working

not planned

not planned

not planned

not planned

not planned

ok

Prop

ok

not
planned

not
planned

not
planned

not
planned

ok

not
planned
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